Author's response to reviews

Title: Study of trunk asymmetry in normal children and adolescent

Authors:

Theodoros B Grivas (grivastb@panafonet.gr)
Elias Vasiliadis (nadel@otenet.gr)
Georgios Koufopoulos (gkfpls@yahoo.com)
Dimitrios Segos (segosd@hotmail.com)
Georgios Triantafilopoulos (geotriantas@ath.forthnet.gr)

Version: 2 Date: 28 September 2006

Author's response to reviews: see over
Dear Sir,

In this cross sectional study, the authors report on the trunk asymmetry in a sample of “normal” Mediterranean school children and adolescents. The preferred position of examination for trunk asymmetry, that is the standing versus the sitting forward bending position, is determined. The incidence of small curves based on scoliometer readings is also reported. The sitting position is recommended as the standard position for examining the rib or loin hump during school screening as it is exposing the real trunk asymmetry.

I hope that this work meets the high standards of your prestigious journal.

Sincerely yours

Dr. Theodoros B Grivas, MD